Author's response to reviews

Title: Effects of Neighborhood Geodemographic Profiles on Healthcare Service Wait Time: A Case Study on Cardiac Care

Authors:

Li Tao (ltao@comp.hkbu.edu.hk)
Jiming Liu (jiming@comp.hkbu.edu.hk)
Bo Xiao (boxiao@comp.hkbu.ddu.hk)

Version: 2 Date: 8 September 2012

Author's response to reviews:

Our revisions in this version of the manuscript are:
1. we declared the "competing interests".
2. we specified the "authors' contribution".
3. we stated that the data used in our study is openly available.